
STDN DAILY REPORT
FOR GMT DAYS

05, 06 AND 07 FEBRUARY 2001

Part I.  Operations

05 FEBRUARY

A. SN Anomalies :

   1. TERRA Support                                    05/1436-1439Z

      TERRA POCC reported having no command upload capability
      from 14:36:47Z until 14:38:47Z. POCC retransmitted the upload
      successfully. No data loss declared reason for anomaly is
      unknown. TTR # 23565

      TDS SSA1F/R 1433-1459Z 2 Min. Svc Loss only

      ANOMALY UNKNOWN
B. ISS/ECOMM Anomalies  -  None.

C. GN ANOMALIES:

   1. SKS/QST Support                                 05/0004-0019Z

      The Bit Sync failed to lock during entire pass resulting in the
      loss of 4K real time data and 262K back orbit Telemetry data.
      The most likely cause is the intermittent Combiner problem which
      is being investigated. TTR # 23567 DR # 18028

      ANOMALY UNKNOWN
      11M 0004-0019Z 15 Min. 14 Sec. Svc/Data Loss (Recov
                                                                                                  unknown)

D.  EUVE Mission Support terminated on 31 January 2001.
       All Mission support requirements fulfilled.



E.  NAM 512 was issued on February 3, 2001.

06 FEBRUARY

A. SN Anomalies  -  None.

B. ISS/ECOMM Anomalies  -  None.

C. GN Anomalies  -  None.

07 FEBRUARY

A. SN Anomalies:

   1 WSGT/EDOS Support                          05/2300-07/1827Z

      WSGT reported at 036/2300Z EDOS began taking hits on there
      50mbps science data being played back from WSGT/GSIF
      (ground station interface facility). Investigation by GE-Americom
      uncovered a “bandit carrier”, a CW signal within the bandwidth of
      transponder 15 (GE-2 satellite) which carries the EDOS 50mbps
      science data. The anomaly is under investigation.  TTR # 23568

      CP ANOMALY

B. ISS/ECOMM Anomalies  -  None.

C. GN Anomalies  -  None.

   1. AGS/QST Support                               07/092430-092431Z

      AGS reported intermittent Doppler readings on the TDF and
      tracking data combiner (Microdyne 1620). The site also indicated
      a loss of lock occurred for 1 second. The problem was traced to
      setup for EO1 (previous support) on the SA930 tracking combiner
      "profile" not being loaded. A new default settings for EO1,



      reloaded QST setup and the receivers 2nd. LO remained in the
      programmed XTAL mode instead of resetting to external mode.
      This QST data hit/Doppler problem only appeared after an EO1
      support, ALL other S/C profiles were tested, found OK. (NO tests
      have been completed using the S/C as a target of
      opportunity to complete the verification of proper operation.)
      TTR # 23571 CDS 18039

      ANOMALY UNKNOWN
      11 Meter 0922-0935Z 1 Sec.Svc/Data Loss Unknown if
      recoverable

   2. SGS/EO-1/LSAT-7 Support                07/084500-121518Z

       The site indicated that all settings on both the analog and digital
       matrix were verified according to profiles. However, the 782 did
       not receive proper signals. The error was located to a faulty
       connection in the analog matrix (782 ST to PTP1 card3). A power
       cycled of the analog matrix was performed to restored service.
       TTR # 23574 CDS # 18048, 18049

       STATION EQUIPMENT
       EO-1   11 Meter 084500-085717Z No Svc/Data Loss
       EO-1   11 Meter 120146-121518Z No Svc/Data Loss
       LSAT-7 11 Meter 102203-103628Z No Svc/Data Loss

   3. SGS/QST/LSAT-7 Supports               07/133752-162300Z

       The PTP cards showed that the clock was absent, power cycled
       the PTP. Still no clock on PTP1/Card 3. Changed PTP1#3 with
       an identical card in our backup PTP. Power up sequence showed
       good clock on all cards.  TTR # 23575  CDS 18050

      SATATION EQUIPMENT
      LSAT-7 11 Meter 133753-135214Z 14 Mins 22 Secs Service
      Loss QST 11 Meter 1608-1623Z 15 Mins Svc/Data loss Unknown
      if recoverable.

   4. AGS/FSAT Support                              07/2240-2259Z



       AGS reported the FTP of data to the project was delayed due to
       DB1 timeout on Pac#3. PB from tape required to Pac#3 after
       reboot of Pac#3. No loss of data to project.
       TTR # 23576  CDS 18051

        ANOMALY UNKNWON
       TOTS 222807-223659Z No Service/Data loss

D. EUVE scheduled 1 event at 211111-220111Z to turn the satellite
     transponder off.

E. STS-98 LIFTED OFF FROM  KSC AT 038/2313:01.954

Part II . Testing Anomalies

A. SN Test  -  None.

B. GN Test  -  None.

Part III.  Equipment Status Changes  -  None.

$  = Changed ETRO
** = New Items

Part IV. Scheduled Activities:

MRTS-1 AND MRTS-2 TDRS Test                             13/2130-08/0200Z

Aqua Flight Operations Team (FOT) Mission Test       08/1430-0300Z

AGS/SGS/WGS TERRA GSIP Parallel Operations     08/1500-1513Z
Phase 3 Test

TITAN CENTUAR Playback Scheduled                           08/1500-1700Z



Part V.  Launch Forecast Changes

* 1.) D5907LS  (TITAN-IV/B41)          INDEFINITE .,2001  T-0 =
         LAUNCH PERIOD


